Author’s response to reviews

Title: Sleep disturbances in an arctic population: The Tromso study

Authors:

Arne Fetveit (arne.fetveit@medisin.uio.no)
Jorund Straand (jorund.straand@medisin.uio.no)
Bjorn Bjorvatn (bjorn.bjorvatn@isf.uib.no)

Version: 3 Date: 26 May 2008

Author’s response to reviews:

Dear Editor,

We have now edited our manuscript according to the formatting requirements:

1. The text "Study protocol" is removed from the title page.
2. Box 1 is converted to Table 1, and references to tables 1, 2 and 3 are up-dated.
3. All tables are converted to fit into an A4 portrait page.

Best regards
Arne Fetveit